Author’s response to reviews

Title: Antitumor effects of Quercetin on Y79 retinoblastoma cells via JNK and p38 MAPK pathways

Authors:

Ming Zhou (Mance.Heatherington779@hotmail.com)

Haojie Liu (ldqxbr319207@163.com)

Version: 3 Date: 17 Nov 2017

Author’s response to reviews:

1. Please add a short, brief Conclusion section after the Discussion section of your main manuscript, in order to adhere to journal submission requirements.

Response: A short brief conclusion is now included in the paper.

2. Please change the text under Consent to Publish to: "not applicable".

Response: It is now corrected.

3. Thank you for including a data availability statement. This statement should detail the location of the raw data underlying the conclusions made in this study. We note that only summarized data is presented in this manuscript, so we would like to ask that you amend this statement. If there are any ethics restrictions preventing the sharing of the raw data, please state this. If you do not wish to share the data, please include a reason why it will not be shared in this section.

Response: We have now edited the statement indicating that the data presented here is summarized data and the raw data is under ethical restrictions and is hence not included